

G OPEN ACCESS

Citation: Wei X, Zhang Z, Chong MKC, Hicks JP, Gong W, Zou G, et al. (2021) Evaluation of a package of risk-based pharmaceutical and lifestyle interventions in patients with hypertension and/or diabetes in rural China: A pragmatic cluster randomised controlled trial. PLoS Med 18(7): e1003694. https://doi.org/10.1371/journal. pmed.1003694

Academic Editor: David Peiris, The George Institute for Global Health, UNSW Sydney, AUSTRALIA

Received: October 14, 2020

Accepted: June 13, 2021

Published: July 1, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003694

Copyright: © 2021 Wei et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

RESEARCH ARTICLE

Evaluation of a package of risk-based pharmaceutical and lifestyle interventions in patients with hypertension and/or diabetes in rural China: A pragmatic cluster randomised controlled trial

Xiaolin Wei¹, Zhitong Zhang¹, Marc K. C. Chong², Joseph P. Hicks³, Weiwei Gong⁴, Guanyang Zou⁵, Jieming Zhong⁴, John D. Walley³, Ross E. G. Upshur¹, Min Yu⁴*

1 Dalla Lana School of Public Health, University of Toronto, Toronto, Ontario, Canada, 2 School of Public Health and Primary Care, Chinese University of Hong Kong, Hong Kong, China, 3 Nuffield Centre for International Health and Development, University of Leeds, Leeds, United Kingdom, 4 Zhejiang Provincial Centre for Disease Control and Prevention, Hangzhou, China, 5 Guangzhou University of Chinese Medicine, Guangzhou, China

* myu@cdc.zj.cn

Abstract

Background

Primary prevention of cardiovascular disease (CVD) requires adequate control of hypertension and diabetes. We designed and implemented pharmaceutical and healthy lifestyle interventions for patients with diabetes and/or hypertension in rural primary care, and assessed their effectiveness at reducing severe CVD events.

Methods and findings

We used a pragmatic, parallel group, 2-arm, controlled, superiority, cluster trial design. We randomised 67 township hospitals in Zhejiang Province, China, to intervention (34) or control (33). A total of 31,326 participants were recruited, with 15,380 in the intervention arm and 15,946 in the control arm. Participants had no known CVD and were either patients with hypertension and a 10-year CVD risk of 20% or higher, or patients with type 2 diabetes regardless of their CVD risk. The intervention included prescription of a standardised package of medicines, individual advice on lifestyle change, and adherence support. Control was usual hypertension and diabetes care. In both arms, as usual in China, most outpatient drug costs were out of pocket. The primary outcome was severe CVD events, including coronary heart disease and stroke, during 36 months of follow-up, as recorded by the CVD surveillance system. The study was implemented between December 2013 and May 2017. A total of 13,385 (87%) and 14,745 (92%) participated in the intervention and control arms, respectively. Their mean age was 64 years, 51% were women, and 90% were farmers. Of all participants, 64% were diagnosed with hypertension with or without diabetes, and 36% were diagnosed with diabetes only. All township hospitals and participants completed the 36-

Data Availability Statement: According to the ethical agreements, the data is owned by the Zhejiang Provincial CDC, and available for research purpose of the study team. The data cannot be posted and downloaded in a public data depository, and not be able to be transmitted out of China due to the National Privacy Regulation. Only aggregated data can be shared upon request to: Department of Chronic Non-communicable Disease Prevention and Control at Zhejiang Provincial CDC, China (Email: zhejiangcdc_ncd@163.com).

Funding: The study was funded by Department for International Development, United Kingdom (https://www.gov.uk/government/organisations/ department-for-international-development) (funding number: COMDIS-HSD, received by XW and JW). The study also received operating support from the Zhejiang Health Commission, China (https://wsjkw.zj.gov.cn/#, received by MY). XW and RU are also endowed Dalla Lana Chairs supported by the Dalla Lana School of Public Health at the University of Toronto (https://www. dlsph.utoronto.ca). The funders of the study had no role in study design, data collection, data analysis, data interpretation, or writing of the report.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AOR, adjusted odds ratio; CHD, coronary heart disease; CVD, cardiovascular disease; GLMM, generalised linear mixed model; HICs, high-income countries; IRR, incidence rate ratio; ITT, intention-to-treat; LMICs, low- and middle-income countries; OR, odds ratio; SDG, Sustainable Development Goal; Zhejiang CDC, Zhejiang Provincial Centre for Disease Control and Prevention. month follow-up. At 36 months, there were 762 and 874 severe CVD events in the intervention and control arms, respectively, yielding a non-significant effect on CVD incidence rate (1.92 and 2.01 per 100 person-years, respectively; crude incidence rate ratio = 0.90 [95% CI: 0.74, 1.08; P = 0.259]). We observed significant, but small, differences in the change from baseline to follow-up for systolic blood pressure (-1.44 mm Hg [95% CI: -2.26, -0.62; P < 0.001]) and diastolic blood pressure (-1.29 mm Hg [95% CI: -1.77, -0.80; P < 0.001]) in the intervention arm compared to the control arm. Self-reported adherence to recommended medicines was significantly higher in the intervention arm compared with the control arm at 36 months. No safety concerns were identified. Main study limitations include all participants being informed about their high CVD risk at baseline, non-blinding of participants, and the relatively short follow-up period available for judging potential changes in rates of CVD events.

Conclusions

The comprehensive package of pharmaceutical and healthy lifestyle interventions did not reduce severe CVD events over 36 months. Improving health system factors such as universal coverage for the cost of essential medicines is required for successful risk-based CVD prevention programmes.

Trial registration

ISRCTN registry ISRCTN58988083.

Author summary

Why was this study done?

- Pharmaceutical interventions have been shown to reduce cardiovascular disease (CVD) events, but the evidence has limited policy implications as medicines in trials have been typically been provided free of charge.
- Four previous trials examining CVD-risk-based management treatments in highincome countries were focused on healthy lifestyle interventions and did not identify any clear effects on CVD events. Two previous trials in low- and middle-income countries implemented both pharmaceutical and lifestyle interventions, but they were not designed to examine impacts on CVD events.

What did the researchers do and find?

• We conducted a cluster randomised controlled trial in rural China among 31,326 patients with (1) hypertension who had a CVD risk of 20% or higher or (2) diabetes, and followed participants up for 36 months. Interventions included prescription of a standardised package of medicines, individual advice on lifestyle change, and adherence support provided by a team led by family doctors. We compared this to usual existing care.

- The study found that our comprehensive pharmaceutical and lifestyle interventions did not reduce severe CVD events.
- We observed significant, but small, differences in the change from baseline to follow-up of systolic blood pressure (-1.44 mm Hg) and diastolic blood pressure (-1.29 mm Hg) in the intervention arm compared to the control arm. Self-reported adherence to recommended medicines was significantly higher in the intervention arm compared with the control arm at 36 months.
- In the process evaluation, we identified that the unaffordability of the medications, caused by lack of health insurance coverage and the non-availability of cheaper generic medicine, appeared to discourage patients from taking recommended medicines.

What do these findings mean?

- Comprehensive pharmaceutical and lifestyle interventions were feasible and effective in improving medicine uptake and reducing blood pressure, but not in achieving a reduction in severe CVD events, when implemented in a setting where the costs of essential medicines were insufficiently covered.
- This study highlights the importance of improving universal health coverage as a prerequisite for effective CVD prevention programmes in primary care in similar settings.

Introduction

Cardiovascular disease (CVD) is the world's leading cause of mortality, representing 31% of all global deaths in 2016 [1]. The United Nations Sustainable Development Goal (SDG) 3 calls for a one-third reduction in CVD deaths by 2030. However, this goal will not be achieved without progressive prevention activities that include adequate control of blood pressure, glucose, and lipid levels, as well as smoking cessation, and reducing salt, alcohol, sugar, and trans fat intake [2]. Primary care provides the best setting for these solutions because hypertension and diabetes, the 2 most common risk factors for CVD, are managed by family doctors, who can provide continuous, coordinated, and comprehensive care. However, hypertension and diabetes are often treated and managed separately, as reflected in the frequent existence of separate guide-lines and programmes. In addition, the 2 diseases are both typically inadequately controlled in low- and middle-income countries (LMICs) compared to high-income countries (HICs). Recent multi-country studies reported that only 10% of patients with hypertension had controlled hypertension in LMICs, compared to 50% in HICs [3,4].

To achieve SDG 3, a comprehensive treatment approach consisting of both pharmaceutical and lifestyle strategies is urgently needed in LMICs. The ability of recommended antihypertensive medicines, aspirin, and statins to prevent CVD events has been well documented in trials [1]. Recommended pharmaceutical therapy, often in the form of a combined polypill, can reduce CVD risk by 50%–60% in the long term [5], but the effects are often more modest, as shown in the PolyIran trial [6] (34% reduction in hazard ratio for major CVD events [95% CI: 20%, 45%] over 60 months). Meta-analyses of healthy lifestyle interventions, including those targeting weight loss, alcohol reduction, and smoking cessation, have shown moderate

improvements in biomarkers of CVD risk factors, such as blood pressure, body mass index (BMI), and total serum cholesterol [7]. Thus, combined pharmaceutical and lifestyle interventions could be more beneficial to patients at high risk of CVD. Several trials have reported positive effects of combined pharmaceutical and lifestyle interventions, including reduced patient CVD risks, blood pressure, and lipid profiles, but none demonstrated reductions in CVD events [8–11]. The strategy of risk-based pharmaceutical and lifestyle interventions has been suggested in national clinical guidelines; however, the policy question of whether this strategy has any impact to achieve SDG 3 remains unanswered due to insufficient evidence [12].

Our previous study identified health system barriers to sustainable CVD risk management, such as insufficient knowledge amongst family doctors, use of ineffective medicines, and lack of treatment support [13]. China's recent health reforms provided an opportunity for better control of non-communicable diseases as they strengthened primary care capacities and improved hypertension and diabetes management [14]. Based on this, we designed and implemented a comprehensive package of pharmaceutical and healthy lifestyle interventions embedded into routine primary care practice for primary prevention of CVD among patients with diabetes and hypertension. We have previously reported feasibility results from our pilot study [15], the trial protocol [16], and the effect on pharmaceutical management [17]. Here we report the evaluation of relevant interventions against the trial's primary and secondary outcomes during 36 months of follow-up.

Methods

Study design

Our study is a pragmatic, parallel group, 2-arm, cluster randomised, controlled, superiority trial. It aims to assess the effectiveness of a package of comprehensive pharmaceutical and healthy lifestyle interventions targeting primary prevention of CVD among patients with high CVD risk in rural China. We conducted the research in township hospitals, which are rural primary care facilities. Township hospitals and their catchment populations are the clusters of the trial. Details of the trial design and analysis plan have been published elsewhere [16]. In brief, we selected all township hospitals, except 1 that served as the pilot site, in 3 counties in Zhejiang Province. Cluster eligibility criteria included having electronic medical records covering the last 2 years for residents in these townships. This study was approved by the Ethics Review Board of the University of Leeds, UK (reference HSLTLM/12/010) and the Ethics Committee of Zhejiang Provincial Centre for Disease Control and Prevention, China (reference 18/06/2012). Written informed consent was obtained from all participating individuals. We used the CONSORT cluster trials checklist (S1 CONSORT Checklist) to help in reporting the study [18].

Participants

We recruited participants from township hospitals based on collected health records from all residents in their catchment areas. Patients were considered eligible if they had the following characteristics: (1) aged 50 to 74 years; (2) permanent residents in the township; (3) having either hypertension with a 10-year CVD risk \geq 20% calculated using the Asian equation [19], or type 2 diabetes (with or without hypertension); (4) free from any diagnosed mental diseases or physical disabilities, any history of severe CVD events, or other diseases as defined in the protocol [16]; (5) not currently hospitalised or living in a long-term care facility; (6) no serious adverse effects to the recommended medicines, and (7) having a diastolic blood pressure \geq 60 mm Hg. After cluster randomisation, all potentially eligible participants were invited to visit the township hospital, where family doctors explained the study, recruited patients, and obtained their consent.

Randomisation and masking

In December 2013, we randomised a total of 67 township hospitals, without stratification, to the intervention and control arms in a 34:33 ratio, via an independent statistician using computer-generated random numbers. All eligible consenting participants received either the intervention or control treatment based on the treatment allocated to their township hospital. Due to the nature of the interventions, health providers and patients could not be blinded, but the analysis of the trial was blinded.

Procedures

In both arms patients had recurring booked consultations with their doctors at least quarterly. All recommended medicines were available in township hospitals or pharmacies in both arms. Patients purchased their medications from township hospitals or pharmacies using prescriptions from their family doctors, and then reported their adherence to the prescriptions in the next consultation.

In the control arm, usual care of hypertension and diabetes continued per existing practice. No additional recommendations for pharmaceutical or healthy lifestyle modifications were provided. The 2010 Chinese national hypertension [20] and diabetes [21] guidelines were available, but they were often not referred to in primary care consultations because the guide-lines contain large amounts of information for tertiary care [22]. Family doctors treated hypertension and diabetes according to their own discretion and group practice. No specific training regarding CVD risk management was provided. Healthy lifestyle change education was given based on controlling either hypertension or diabetes, but compared to the intervention arm this was not using a holistic approach to CVD risk reduction. Treatment adherence support was not provided.

In the intervention arm, patients received prescriptions and healthy lifestyle education for treatment of their existing medical conditions and CVD prevention. All patients with hypertension, or hypertension and diabetes, were prescribed a standard combination [23] of (1) 2 antihypertensives (2 different kinds selected from thiazide diuretics, calcium channel blockers, angiotensin-converting enzyme inhibitors, angiotensin II receptor blockers, and beta-blockers) [24], (2) a statin, and (3) a low dose aspirin, unless contraindicated. Patients with only diabetes received the same pharmaceutical package but with only 1 antihypertensive, plus their anti-diabetic medicines, if any. Patients already on other older antihypertensive medicines were advised to switch to the recommended medicines in standardised packages. Patients who had a history of, or showed signs of, gastrointestinal bleeding were not prescribed aspirin. During consultation, family doctors provided individualised health education focusing on smoking cessation, salt reduction, and reduction of alcohol consumption depending on the patient's situation. A family treatment supporter was selected at the patient's home to support him/her taking medicines and adhering to lifestyle changes. At the facility level, we provided annual training to all family doctors using our CVD risk management guidelines, which covered (1) prescribing for primary prevention of CVD, (2) personalised advice for healthy lifestyle changes, and (3) advice to improve adherence [16]. Township hospitals held monthly meetings to discuss doctors' experience of using the guidelines and chronic disease management (Table 1).

Data collection

On enrolment, family doctors measured participants' blood pressure using a standardised mercury sphygmomanometer after 5 minutes of seated rest. They also measured their body weight and height, and asked for details on their medication history. These measures were

Target	Intervention strategies in the intervention arm	Usual care in the control arm
Family doctors	Operational guidelines: The guidelines contained a definition of high risk of CVD, workflow chart for patient management, how to diagnose hypertension and type 2 diabetes, suggested prescriptions and CVD risk management practices, lifestyle evaluation and recommendations, treatment support, and managing the side effects of medicines. In addition, we also incorporated elements of China's national hypertension and type 2 diabetes guidelines for township hospitals. Training workshops: Training workshops were provided on an annual basis by local senior doctors based on the operational guidelines, including hypertension/diabetes management, prevention, and communication skills, using lectures, health education videos, case discussions, role plays, and question and answer sessions. Monthly meetings in township hospitals for performance monitoring: A senior doctor led the review of the implementation of intervention strategies, and provided comments to related questions on CVD risk management and treatment of patients by doctors.	 Guidelines and training: National hypertension and type 2 diabetes guidelines were available but were not referred to in practice because the guidelines focus on tertiary care. No specific training/guidelines were given on reducing CVD risks based on a holistic approach. Training workshops: Routine annual training workshops on hypertension and type 2 diabetes management were provided, but there was no research-driven systematic training provided on CVD risk management. Monthly meetings: Routine monthly internal meetings were held, but there were no specific discussions dedicated to CVD risk management.
Patients and their treatment supporters	Recommended medicines: A combination of 2 antihypertensives, a statin, and a low dose of aspirin were recommended to patients with hypertension. One antihypertensive, a statin, and a low dose of aspirin were recommended for patients with type 2 diabetes but not hypertension. Anti-diabetic medicine or insulin was given for diabetic patients if necessary. Health education messages from family doctors during regular follow-up consultations (from monthly to quarterly): In their consultations, family doctors gave targeted health education messages to participants based on their health conditions, e.g., smoking cessation messages to smokers, and messages regarding the benefits of the combined medicines to those who did not adhere to prescriptions. Enhanced follow-up appointment: Participants attended follow-up consultations in township hospitals at least quarterly with their family doctors. Nurses communicated with participants monthly through phone calls/SMS messages. Treatment supporter: Family doctors or nurses helped to identify and train a treatment supporter (normally a family member) who reminded patients to take medicines, maintain a healthy lifestyle, and adhere to their follow-up consultations.	Recommend medications and health education: Treatment and lifestyle changes were offered according to existing knowledge and at the individual clinician's discretion, but based on either hypertension or type 2 diabetes instead of a holistic approach for CVD prevention. In practice, most doctors prescribed only 1 antihypertensive for hypertensive patients, and no antihypertensive for diabetic patients. Very few patients were given a statin or aspirin. Anti-diabetic medicine or insulin was given for diabetic patients if necessary. Existing general health education messages were provided. Follow-up appointment: Patients with hypertension or type 2 diabetes were followed up at least quarterly by family doctors according to national guidelines. Treatment supporter: The option of a treatment supporter was not provided.
Medicines	Modern medicines were available in both the intervention and control ar	ms, and their prices were the same in the 2 arms.

Table 1. Intervention strategies to reduce the risk of cardiovascular disease (CVD).

https://doi.org/10.1371/journal.pmed.1003694.t001

repeated on a quarterly basis when participants were followed up in township hospitals. All information was recorded in an internet-based public health management information system. The trial stopped 36 months (May 2017) after enrolment of the last participant based on our protocol.

All acute severe CVD events were collected through Zhejiang CDC's surveillance system [25] based on the adapted WHO MONICA definitions [26] compatible with ICD-10 codes. These included coronary heart disease (CHD) and stroke, where CHD includes acute myocardial infarction, ischaemic cardiac arrest, and unclassifiable CHD deaths, and stroke includes haemorrhagic stroke (e.g., subarachnoid and intracerebral), ischaemic stroke/infarction (e.g., thrombosis and embolism), and unclassifiable stroke [27]. We did not include minor CVD events such as angina and transient ischaemic attacks because they are often unrecorded. To be eligible as endpoints, all CVD events had to be verified from hospital records, and were reviewed and verified by centres for disease control at the county and provincial levels. All reported CVD events were then also been verified by primary care facilities through home visits. The process was supervised under the Validation Committee in the Zhejiang CDC. We previously reported on the data collection, reporting, and validation of the Zhejiang CDC's surveillance system [25]. In addition, family doctors asked participants if they had experienced any stroke or heart attack events that were diagnosed by hospitals at the 12th, 24th, and 36th month following randomisation; if any, the CVD events were reported to the Validation Committee and then verified and included in the records. Participants were classified as lost to follow-up if they could not be contacted after 3 attempts (via telephone, message, or home visit) and their CVD status was unknown by the 36th month.

Outcomes

All outcomes were measured at the patient level. The primary outcome was the number of severe CVD events recorded by Zhejiang's CVD surveillance system [25]. We also collected the following secondary outcomes: (1) mortality due to severe CVD event, (2) number of CHD events, (3) mortality due to CHD event, (4) number of stroke events, (5) mortality due to stroke, (6) time to the first reported severe CVD event, (7) time to mortality due to CVD, (8) change in diastolic and systolic blood pressure (mm Hg) between baseline and 36 months, (9) adherence to the final quarterly follow-up consultation at 36 months, and (10) self-reported adherence at 36 months to 2 antihypertensive drugs for patients with hypertension or 1 antihypertensive drug for patients with only diabetes, aspirin, and statin. We also recorded any minor or serious adverse events. We added secondary outcomes 2–5 to identify any changes in CVD subgroups. In subsequent papers, we will report our other protocol-defined secondary outcomes from our panel data and cost-effectiveness analysis, which are based on different datasets as per our protocol [16]. We also previously reported feasibility measures in our process evaluation paper [28].

Statistical analysis

We calculated the necessary sample size as 32 clusters per arm and 450 patients per cluster, to have 90% power to detect a 20% reduction in the severe CVD event incidence rate after 36 months, based on an assumed CVD event rate of 5% in the control arm and a coefficient of variation of 0.15, with hypothesis testing using a 2-sided *P* value with a 5% threshold for statistical significance.

We calculated the exact follow-up time for each participant until the end of the trial or death. We estimated crude and covariate-adjusted ("adjusted") estimates of the treatment effect on all outcomes using generalised linear mixed models (GLMMs). To estimate the crude treatment effect (as an incidence rate ratio [IRR]) on our primary outcome, we used a GLMM with a fixed effect for treatment arm, a random intercept for cluster, and an "offset" variable for the log of person follow-up time, which used a log-link with Poisson errors. To estimate the adjusted treatment effect, we repeated the same model but included patient-level variables as per table footnotes. We also estimated crude and adjusted treatment effects for all our secondary outcomes using a variety of mixed-effects models, with the same random and fixed effects (unless stated: see relevant table footnotes) included. We employed GLMMs with Poisson errors and log-links (if no evidence of overdispersion) or GLMMs with negative binomial distributions and log-links (with treatment effects estimated as IRRs) to analyse event and mortality outcomes, Cox proportional hazard mixed-effects models (with treatment effects estimated as HRs) for time to event outcomes, GLMMs with normal errors and identity links (with treatment effects estimated as mean differences) for blood pressure outcomes, and GLMMs with binomial errors and logistic links (with treatment effects estimated as odds ratios [ORs]) for adherence and adverse event outcomes. The intra-cluster correlation coefficient for the primary outcome was calculated by dividing the between-cluster variance by the total variance. The variance terms were estimated by fitting the relevant (Poisson/negative binomial) unconditional mixed-effects model with a random intercept for cluster [29,30]. All 95%

confidence intervals and *P* values for GLMMs were based on either the Wald statistic (nonnormal errors) or the *t* statistic (normal errors). We analysed all data using SAS statistical software (PROC MIXED and PROC GLIMMIX functions). Where any patients were missing outcomes or covariates in adjusted analyses, we excluded these patients and performed complete case analyses. We also ran cluster-level analyses for all our outcomes, as reported in <u>S1 Text</u>.

Our primary analysis was of the intention-to-treat (ITT) population and included all participants within all clusters as originally recruited and randomised, including all participants who left the trial after their first consultation or moved out of their cluster catchment area, and those who were wrongly classified as eligible: They all remained in the province so their CVD status was still tracked by the surveillance system. We also defined a "modified ITT" analysis population, which included all clusters and patients as originally recruited and randomised but excluding any patients with missing outcome or covariate data. We also repeated our analyses for all outcomes reported here in a per-protocol analysis population using the same methods described above. This per-protocol analysis population included all clusters as originally recruited and randomised, and across both arms it included all eligible patients except those patients who left the trial within the first quarter or those who were wrongly recruited. In addition, in the intervention arm this population excluded patients who did not have at least 50% adherence to their prescriptions across all their follow-up consultations. This included taking recommended antihypertensives (specifically, patients with hypertension taking a standard combination of 2 antihypertensives, and patients with diabetes only taking 1 antihypertensive) and statins or aspirin. These per-protocol analyses were designed after the trial because we observed a low rate of adherence to prescriptions, beyond what we expected.

We did a range of exploratory subgroup analyses of the primary outcome based on the prerandomisation baseline characteristics of diagnosis (hypertension or hypertension and diabetes versus diabetes only), patient income level (<10,000 RMB [<US\$1,644] versus \geq 10,000 RMB [\geq US\$1,644] per annum), and education (primary school or below versus high school or above). In our analyses of secondary outcomes and in our per-protocol and subgroup analyses, we did not adjust the *P* values or confidence intervals for multiple testing, and we treat the results as exploratory.

Results

We randomised 34 township hospitals to the intervention arm and 33 township hospitals to the control arm. Across all hospitals' catchment areas, based on electronic health records, we identified a total of 31,326 potentially eligible participants: 15,380 in the intervention arm and 15,946 in the control arm. At the recruitment stage (December 2013 to May 2014), 1,995 and 1,201 eligible participants declined to participate in the intervention and control arms, respectively, resulting in recruitment of 13,385 (87%) and 14,745 (92%) participants in the intervention and control arms, respectively (Fig 1).

As shown in Table 2, participants in the 2 arms were similar with respect to baseline characteristics. The mean age was 64 years, 51% were women, 90% were farmers, and 77% had a primary education or below. The median annual per capita income was US\$1,424, reflecting the poor and rural setting. Of all participants, 64% were diagnosed with hypertension with or without diabetes, and 36% were diagnosed with diabetes only. Following their first consultation, 1,144 and 1,694 participants in the intervention and control arms, respectively, chose to stop participating, and 76 and 83 participants in the intervention and control arms, respectively, were found to have pre-existing CVD (and therefore ineligible), but all were included in the ITT analysis population. During the follow-up there were 366 and 483 all-cause mortalities, and 135 and 194 participants moved out of their cluster catchment areas, in the intervention and control



Fig 1. Trial profile. ^aIneligible patients included 2,419,058 who didn't meet the eligibility criteria (aged 50–74 years, with a calculated 10-year CVD risk \geq 20% and having hypertension, or diagnosed with type 2 diabetes); 222 who had mental diseases; 492 who had cancers; 2,076 who had preexisting acute coronary heart disease or had experienced a stroke according to heath records; 293 whose diastolic blood pressure was <60 mm Hg; 16,562 who were out of town for more than 3 months or were non-locals/urban residents; and 1,155 who couldn't take the recommended medicines due to other severe diseases. ^bPatients whose records of acute cardiovascular diseases or death could not be obtained at the end of the 36-month follow-up period were categorised as lost to follow-up. ^cPatients who died (intervention = 366, control = 483), moved out of cluster catchment areas (intervention = 135, control = 194), were subsequently found to have pre-existing acute cardiovascular diseases not identified through initial health records (intervention = 7,6, control = 83), or subsequently decided to leave the trial within the first quarter of the follow-up period (intervention = 1,144, control = 1,694) are all included in the main (covariate unadjusted) intention-to-treat (ITT) analyses. ^dPatients excluded from the per-protocol analysis: (1) patients in the intervention group who did not report taking both suggested hypertensive medicines and a statin/ aspirin for more than half of the follow-up period before the occurrence of acute cardiovascular disease, if they had an occurrence, or until the end of follow-up, if they did not (*n* = 11,523); (2) patients who were subsequently found to have pre-existing acute cardiovascular diseases. It is assess not identified through initial health records (intervention = 7,6, control = 83); and (3) patients who subsequently decide to leave the trial within the first quarter of the follow-up cord (intervention = 1,144, control = 1,694).

https://doi.org/10.1371/journal.pmed.1003694.g001

arms, respectively (Fig_1). All clusters and participants were successfully followed up based on electronic medical records and/or CVD events reported via family doctors, allowing us to have complete data for the primary and secondary outcomes for the ITT population. However, there were some missing data for the adherence and blood pressure outcomes, and for the covariates in adjusted analyses. Missing case details are provided in the footnotes of the results tables.

Our interventions did not have a statistically significant effect on the primary outcome. Among all participants, 762 and 874 severe CVD events were reported in the intervention and

Patient characteristics	Intervention group (<i>n</i> = 13,385)	Control group ($n = 14,745$)	Total (<i>n</i> = 28,130)
Number of clusters	34	33	67
Age (years), mean (SD)	64.3 (6.3)	64.2 (6.1)	64.3 (6.1)
Sex, <i>n</i> (%)			
Male	6,443 (48.1%)	7,316 (49.6%)	13,759 (48.9%)
Female	6,942 (51.9%)	7,429 (50.4%)	14,371 (51.1%)
Body mass index (kg/m ²), mean (SD)	24.0 (3.0)	23.9 (2.9)	23.9 (2.9)
Occupation, <i>n</i> (%)			
Farmer	11,735 (89.9%)	12,960 (90.2%)	24,695 (90.0%)
Worker	205 (1.6%)	181 (1.3%)	386 (1.4%)
Technician	37 (0.3%)	55 (0.4%)	92 (0.3%)
Administrative staff	18 (0.1%)	8 (0.1%)	26 (0.1%)
Service worker	58 (0.4%)	56 (0.4%)	114 (0.4%)
Personal business	197 (1.5%)	165 (1.2%)	362 (1.3%)
Retired	105 (0.8%)	169 (1.2%)	274 (1.0%)
Other	702 (5.4%)	778 (5.4%)	1,480 (5.4%)
Educational level, <i>n</i> (%)			
Primary school or below	10,356 (77.7%)	11,070 (75.4%)	21,426 (76.5%)
High school	2,691 (20.2%)	3,234 (22.0%)	5,925 (21.1%)
College or above	290 (2.1%)	382 (2.6%)	672 (2.4%)
Marital status, n (%)			
Married	11,813 (95.4%)	13,109 (95.6%)	24,922 (95.5%)
Single, divorced, or widowed	566 (4.6%)	610 (4.4%)	1,176 (4.5%)
Annual per capita income (US dollars), median (IQR)	1,424 (712, 1,994) ^a	1,424 (712, 2,374) ^a	1,424 (712, 2,136) ^a
Diagnosis, n (%)			
Hypertension with/without type 2 diabetes	8,800 (65.7%)	9,224 (62.6%)	18,024 (64.1%)
Type 2 diabetes without hypertension	4,585 (34.2%)	5,521 (37.4%)	10,106 (35.9%)
10-year CVD risk in patients with hypertension, median (IQR)	26.5% (23.0%, 30.8%)	26.3% (22.9%, 30.8%)	26.4% (22.9%, 30.8%)

Table 2.	Characteristics	of the	28,130	participants	in the trial.
----------	-----------------	--------	--------	--------------	---------------

^aThe recommended package of medicines cost US\$35–US\$44 per month in the local setting, i.e., US\$420–US\$528 per year.

https://doi.org/10.1371/journal.pmed.1003694.t002

control arms, respectively. The majority (99.4%) of CVD events were collected through Zhejiang's CVD surveillance system, while 4 and 5 CVD events from the intervention and control arms, respectively, were collected via family doctors at their consultations. The observed incidence rate for severe CVD events was 1.92 and 2.01 per 100 person-years in the intervention and control arms respectively, and the crude IRR was 0.90 (95% CI: 0.74, 1.08; P = 0.259; Table 3). The primary outcome's IRR was not substantially altered when adjusting for covariates (Table 3). There was no significant treatment effect on any of the secondary event, mortality, or time to event outcomes, including CVD mortality (Tables 3 and 4). The results from our adjusted analyses were consistent with the crude results for all secondary outcomes (Tables 3 and 4). The cluster-level analyses of both primary and secondary outcomes produced results consistent with those of the patient-level analyses (Tables A–C in S1 Text).

Our per-protocol population included just 642 participants in the intervention arm and excluded 2 clusters in the intervention arm that had no remaining participants, while retaining 12,968 patients and all clusters in the control arm. There were no significant differences between the 2 arms regarding severe CVD events (1.58 versus 1.94 per 100 person-years in the intervention and control groups, respectively; crude IRR = 0.81 [95% CI: 0.55, 1.20;

Outcome	Intervention		Control		Crude IRR (95% CI) ^b	P value ^D	Adjusted IRR (95% CI) ^c	P value ^c
	$N(\%)^{\mathrm{a}}$	Event rate	N (%) ^a	Event rate				
CVD event ^d	762 (5.69%)	1.92	874 (5.93%)	2.01	0.90 (0.74 to 1.08)	0.259	0.87 (0.73 to 1.05)	0.148
CVD mortality	165 (1.23%)	0.42	185 (1.25%)	0.42	0.91 (0.69 to 1.21)	0.516	0.90 (0.68 to 1.18)	0.455
CHD event	92 (0.69%)	0.23	120 (0.81%)	0.28	0.84 (0.63 to 1.11)	0.217	0.86 (0.65 to 1.14)	0.290
CHD mortality	43 (0.32%)	0.11	55 (0.37%)	0.13	0.80 (0.46 to 1.37)	0.410	0.73 (0.39 to 1.37)	0.331
Stroke event	659 (4.92%)	1.66	738 (5.01%)	1.69	0.91 (0.75 to 1.11)	0.359	0.88 (0.73 to 1.07)	0.200
Stroke mortality	122 (0.91%)	0.31	132 (0.90%)	0.30	0.99 (0.75 to 1.30)	0.932	0.99 (0.77 to 1.28)	0.937

Table 3. Comparison of primary and secondary outcomes for disease event rates between the intervention and control arms (individual-level data analyses).

CHD, coronary heart disease; CVD, cardiovascular disease; IRR, incidence rate ratio. Intervention, n = 13,385; control, n = 14,745. The total person-years of follow-up were 39,657 and 43,557 for the intervention group and control group, respectively.

^aIndividual-level summary statistics: *N* is the number of events; percent is the total number of events/total number of individuals (per arm). Event rate is per 100 personyears. For all disease event outcomes (as opposed to disease mortality outcomes), individuals may experience >1 event, and so the N (%) statistics are based on the occurrence/non-occurrence of the event per person only.

^bThe intervention minus control crude IRRs were obtained via a Poisson mixed-effects model (primary outcome) or a Poisson or negative binomial mixed-effects model (secondary outcomes: depending on the level of dispersion) with a single fixed effect of treatment arm and a random intercept for cluster. The 95% CIs and *P* values were calculated based on the Wald statistic. In all crude analyses there were no missing patients.

^cThe intervention minus control adjusted IRRs were obtained via a Poisson mixed-effects model (primary outcome) or a Poisson or negative binomial mixed-effects model (secondary outcomes: depending on the level of dispersion) with fixed effects for treatment arm, patient age, sex, body mass index, occupation (farmer or non-farmer), educational level (primary school or below, or high school or above), marital status (married or non-married), income (less than 10,000 RMB or 10,000 RMB or above), and diagnosis (hypertension without type 2 diabetes, type 2 diabetes without hypertension, or hypertension with type 2 diabetes), and a random intercept for cluster. The 95% CIs and *P* values were calculated based on the Wald statistic. In all adjusted analyses, 1,112 (8.3%) and 1,432 (9.7%) patients in the intervention and control arms, respectively, were excluded due to missing outcome or covariate data.

^dOnly 14 and 19 patients in the intervention and control arms, respectively, reported a second severe CVD event, and none reported more than 2 severe CVD events. The intra-cluster correlation coefficient for the primary CVD event outcome was 0.024, and was calculated by dividing the between-cluster variance by the total variance. The variance terms were estimated by fitting the relevant (Poisson or negative binomial) unconditional mixed-effects model with only a random intercept for cluster.

https://doi.org/10.1371/journal.pmed.1003694.t003

P = 0.298]). When analysing the per-protocol population there were also no significant results in our secondary outcomes of mortality and events (Tables D–F in <u>S1 Text</u>).

Outcome	Mean (SD) number of days ^a		Crude HR (95% CI) ^b	P value ^b	Adjusted HR (95% CI) ^c	P value ^c
	Intervention	Control				
Time to the first CVD event	554.2 (311.4)	554.8 (314.8)	0.87 (0.72 to 1.05)	0.140	0.86 (0.72 to 1.03)	0.108
Time to CVD mortality	614.8 (314.8)	627.4 (307.2)	0.92 (0.69 to 1.22)	0.543	0.91 (0.69 to 1.19)	0.478

Table 4. Comparison of time to first CVD event and time to CVD mortality outcomes between the intervention and control arms (individual-level data analyses).

CVD, cardiovascular disease; HR, hazard ratio. Intervention, n = 13,385; control, n = 14,745.

^aIndividual-level summary data are presented as mean (SD) number of days until the outcome, excluding censored outcomes.

^bThe intervention minus control crude HRs were obtained using a mixed-effects Cox proportional hazard model with a single fixed effect of treatment arm and a random intercept for cluster. The 95% CIs and *P* values were calculated based on the Wald statistic. In both crude analyses there were no missing patients. ^cThe intervention minus control adjusted HRs were obtained using a mixed-effects Cox proportional hazard model with fixed effects for treatment arm, patient age, sex, body mass index, occupation (farmer or non-farmer), educational level (primary school or below, or high school or above), marital status (married or non-married), income (less than 10,000 RMB or 10,000 RMB or above), and diagnosis (hypertension without type 2 diabetes, type 2 diabetes without hypertension, or hypertension with type 2 diabetes), and a random intercept for cluster. The 95% CIs and *P* values were calculated based on the Wald statistic. In the adjusted analysis of the time to first CVD event outcome, 1,153 (8.6%) and 1,470 (10.0%) patients in the intervention and control arms, respectively, were excluded due to missing covariate data, and in the adjusted analysis of the time to CVD mortality outcome, 1,644 (12.3%) and 2,037 (13.8%) patients in the intervention and control arms, respectively, were excluded due to missing covariate data.

https://doi.org/10.1371/journal.pmed.1003694.t004

Outcome	itcome Mean (SD) blood pressure (mm Hg)		ood pressure (mm Crude mean difference (95% CI) ^a		Adjusted mean difference (95% CI) ^b	P value ^b
	Intervention	Control				
Systolic bloo	d pressure					
Baseline	136.2 (11.7)	135.9 (11.4)				
Endline	132.4 (9.34)	133.4 (9.71)	-1.44 (-2.26 to -0.62)	< 0.001	-1.50 (-2.30 to -0.70)	< 0.001
Diastolic blo	od pressure					
Baseline	81.7 (6.81)	81.8 (6.67)				
Endline	79.5 (6.31)	80.6 (6.30)	-1.29 (-1.77 to -0.80)	< 0.001	-1.31 (-1.79 to -0.83)	<0.001

Table 5. Comparison of systolic and diastolic blood pressure outcomes between intervention and control arms (individual-level data analyses).

Intervention, *n* = 13,385; control, *n* = 14,745.

^aThe intervention minus control crude mean endline differences were obtained via a linear mixed-effects model with a single fixed effect of treatment arm and a random intercept for cluster. The 95% CI and *P* values were calculated based on the *t* statistic. In both crude analyses there were no missing patients.

^bThe intervention minus control adjusted mean endline differences were obtained via a linear mixed-effects model with fixed effects for treatment arm, patient age, sex, body mass index, occupation (farmer or non-farmer), educational level (primary school or below, or high school or above), marital status (married or non-married), income (less than 10,000 RMB or 10,000 RMB or above), and diagnosis (hypertension without type 2 diabetes, type 2 diabetes without hypertension, hypertension with type 2 diabetes), and a random intercept for cluster. The 95% CI and *P* values were calculated based on the *t* statistic. In the adjusted analysis of the systolic blood pressure outcome, 2,624 (19.6%) and 3,339 (22.6%) patients in the intervention and control arms, respectively, were excluded due to missing covariate or outcome data, and in the adjusted analysis of the diastolic blood pressure outcome 2,620 (19.6%) and 3,338 (22.6%) patients in the intervention and control arms, respectively, were excluded due to missing covariate or outcome data.

https://doi.org/10.1371/journal.pmed.1003694.t005

We did observe significant crude differences in the intervention arm compared to the control arm in the change from baseline to 36 months regarding systolic blood pressure (-1.44 mm Hg [95% CI: -2.26, -0.62; P < 0.001]) and diastolic blood pressure (-1.29 mm Hg [95% CI: -1.77, -0.80; P < 0.001]), with similar adjusted results (Table 5). In our per-protocol analyses we observed a similarly sized significant difference in diastolic blood pressure change (-1.25 mm Hg [95% CI: -2.02, -0.49; P = 0.001]), but a non-significant difference in systolic blood pressure change (-0.64 mm Hg [95% CI: -1.88, -0.60; P = 0.315]; Table G in S1 Text).

Fig 2 illustrates that adherence to recurring booked quarterly consultations remained high in both arms throughout the trial, only appearing to decline slightly from 4 months until 36 months in both arms. Correspondingly, there was no significant difference between arms in patients' adherence to their final quarterly consultations (adjusted OR [AOR] = 1.44 [95% CI: 0.88, 2.36; P = 0.144]). In total, 84.9% (11,359) of patients in the intervention arm had a treatment supporter, compared with only 0.4% (63) of patients in the control arm (AOR = 159.32, [95% CI: 124.53, 203.81; P < 0.001]). In most cases, treatment supporters were patients' spouses and children. Among patients with hypertension, either alone or with diabetes, as shown in Fig 3A and 3B, only 22.5% reported adherence to 2 antihypertensive drugs at baseline in both arms. This then steadily increased to 37.2% at 36 months in the intervention arm compared to 16.8% in the control arm (AOR = 4.75 [95% CI: 3.82, 5.91; *P* < 0.001]). In the intervention arm, patients reported significant improvement in adherence to aspirin (1% at baseline and 16.7% at 36 months) and statins (0.4% at baseline and 8% at 36 months), compared to consistently low levels of adherence reported in the control arm (aspirin, 1% at baseline and 1.7% at 36 months; AOR = 12.1 [95% CI: 7.69, 18.9; *P* < 0.001]; and statin, 0.4% at baseline and 0.8% at 36 months; AOR = 9.69 [95% CI: 5.65, 16.6; P < 0.001]). Among patients with only diabetes (Fig 3C and 3D), we did not identify a significant difference between the 2 arms in adherence to 1 antihypertensive at 36 months (AOR = 1.31 [95% CI: 0.93, 1.84; P = 0.13]). Their adherence to aspirin was around 1% at baseline in both arms, but this increased substantially in the intervention arm (to 8.9%) but not in the control arm (1.1%),



Fig 2. Adherence to booked quarterly consultations. Adherence to booked consultations is calculated as the proportion of patients in each arm who responded to their quarterly (i.e., every 3 months) follow-up clinical consultations with health workers within each 3-month follow-up period. A mixed-effects logistic regression analysis of patients' adherence (yes/no) to their final quarterly booked consultation—with fixed effects for treatment arm, patient age, sex, body mass index, occupation (farmer or non-farmer), educational level (primary school or below, or high school or above), marital status (married or non-married), income (less than 10,000 RMB or 10,000 RMB or above), and diagnosis (hypertension without type 2 diabetes, type 2 diabetes without hypertension, hypertension with type 2 diabetes), and a random intercept for cluster—demonstrated no significant difference between treatment arms (intervention versus control adjusted odds ratio = 1.44 [95% CI: 0.88, 2.36; P = 0.144]). This analysis excluded 1,112 (8.3%) and 1,432 (9.7%) patients from the intention-to-treat population intervention (n = 13,385) and control (n = 14,745) arms, respectively, due to missing covariate or outcome data.

https://doi.org/10.1371/journal.pmed.1003694.g002

reflecting a significantly higher level of adherence in the intervention arm at 36 months (AOR = 9.09 [95% CI: 5.44, 15.2; P < 0.001]). For these patients, there was a similar pattern observed for statin adherence, with near 0% adherence at baseline in both arms but 6.1% adherence in the intervention arm, compared to 0.4% in the control arm, at 36 months (AOR = 12.8 [95% CI: 6.68, 24.5; P < 0.001]).

We analysed patients' self-reported adherence (yes/no) to at least 1 or 2 antihypertensive medicines, aspirin, and/or a statin at 36 months of follow-up via mixed-effects logistic regression models, with fixed effects for treatment arm, patient age, sex, body mass index, occupation (farmer or non-farmer), educational level (primary school or below, or high school or above), marital status (married or non-married), income (less than 10,000 RMB or 10,000 RMB or above), and diagnosis (hypertension without type 2 diabetes, type 2 diabetes without hypertension, hypertension with type 2 diabetes), and a random intercept for cluster. In the subgroup of patients with hypertension (intervention n = 8,800, control n = 9,224), selfreported adherence to at least 2 antihypertensive drugs, aspirin, and/or a statin at 36 months was significantly higher in intervention arm patients compared to control arm patients (AOR for self-reported adherence to at least 2 antihypertensive medicines at 36 months = 4.75 [95% CI: 3.82, 5.91; P < 0.001]; AOR for self-reported adherence to aspirin at 36 months = 12.1 [95% CI: 7.69, 18.9; P < 0.001]; AOR for self-reported adherence to a statin at 36 months = 9.69 [95% CI: 5.65, 16.6; P < 0.001]). In all these subgroup analyses, 851 (9.7%) and 1,094 (11.9%) patients in the intervention and control arms, respectively, were excluded due to missing covariate or outcome data. In the subgroup of patients with type 2 diabetes only (intervention n = 4,585, control n = 5,521), self-reported adherence to at least 1 antihypertensive medicine at



Fig 3. Self-reported adherence to medicines. (A and B) Self-reported adherence to medicines among patients with hypertension or hypertension and type 2 diabetes in the (A) intervention and (B) control groups. (C and D) Self-reported adherence to medicines among patients with type 2 diabetes only in the (C) intervention and (D) control groups.

https://doi.org/10.1371/journal.pmed.1003694.g003

36 months was not significantly higher in intervention arm patients compared to control arm patients (AOR for self-reported adherence to at least 1 antihypertensive medicine = 1.31 [95% CI: 0.93, 1.84; P = 0.13]), but self-reported adherence to aspirin and/or a statin at 36 months was significantly higher in intervention arm patients compared to control arm patients (AOR for aspirin = 9.09 [95% CI: 5.44, 15.2; P < 0.001]; AOR for statin = 12.8 [95% CI: 6.68, 24.5; P < 0.001]). In all these subgroup analyses, 1,494 (32.6%) and 1,999 (36.2%) patients were excluded from the intervention and control arms, respectively, due to missing covariate or outcome data.

A total of 148 (1.1%) and 185 (1.3%) patients reported minor adverse events in the intervention and control arms, respectively (AOR = 0.927 [95% CI: 0.514, 1.673; P = 0.802]). Of 333 reported adverse events, 134 (40.2%) were gastrointestinal discomfort, 73 (21.9%) were dizziness, 50 (15.0%) were hypodynamia, 35 (10.5%) were undefined bleeding, and 41 (12.4%) were other unclassified conditions. No significant difference was identified for any category of adverse event (Table I in <u>S1 Text</u>). We did not identify any serious adverse events that required emergency medical attention.

Discussion

To our knowledge, this is the first large-scale, pragmatic, cluster randomised controlled trial with long-term follow-up that investigates the effect of a comprehensive CVD-risk-based prevention strategy on CVD events. We did not observe a statistically significant difference regarding the primary outcome measure of severe CVD events within 36 months. Nor did we identify a statistically significant effect on the primary outcome in our per-protocol analyses, where we only included intervention arm participants who reported at least 50% adherence to recommended prescriptions. However, only 4.6% of patients were retained in the intervention arm per-protocol population, despite the relatively low adherence threshold for inclusion. Therefore, along with the limitations of per-protocol analyses, to robustly understand the actual treatment effect in this context, as opposed to the effect of just providing the treatment option, would clearly require further study. There was a significant difference in the secondary outcomes of systolic and diastolic blood pressure change from baseline to follow-up that favoured the intervention over the control treatment, but it was small. Related to this, we also observed statistically significant and clinically moderate to large improvements in patientreported adherence to 2 antihypertensives, aspirin, and statins, albeit the overall adherence rates to all medicines were low. We found no other statistically significant intervention effects for any of our other secondary outcomes within the crude or adjusted ITT or per-protocol population analyses.

Several health system issues may limit the effectiveness of such interventions. First, low health insurance coverage for medicines and high out-of-pocket payments may discourage patients from taking recommended medicines. Though our intervention substantially improved adherence rates (i.e., to 37% for recommended antihypertensives, 17% for aspirin, and 6% for statins in the intervention arm), the rates were nevertheless far lower than those in trials providing free medicines. For example, the PolyIran trial [6] achieved 81% adherence, and an atorvastatin trial [31] had 85% adherence. Based on strategies of motivating community health workers and providing free medicines, the HOPE4 trial reported 84% adherence to antihypertensives and statin [11]. Our adherence rates mimic conditions in real-life settings, where less than 50% of patients with hypertension regularly take antihypertensives [32]. In our setting, the recommended package of medicines cost US\$35-US\$44 per month, with participants having a median annual per capita income of US\$1,424. Our process evaluation revealed that affordability of medicines was a major challenge due to low health insurance coverage for outpatient costs. In practice, most patients paid their outpatient medicines out of pocket [28,33]. China extended universal health coverage to everyone in 2009, which reduced the proportion of out-of-pocket health expenditure, but not the absolute level [34]. In addition, outpatient costs, as often seen in primary care, are proportionally much less covered than inpatient costs. This problem is exacerbated as cheaper generic medicines often become unavailable in township hospitals because the companies supplying them cannot make profits under current bidding regulations [28]. This affordability challenge has created a health equity crisis in primary care in China, wherein the poorest quintile of patients with diabetes experience the highest burden of catastrophic health expenditure [35]. Second, our participants were mainly older people who had previously been treated for hypertension or diabetes, and they were commonly reluctant to change from their existing combination tablet, which contained antihypertensive and traditional Chinese medicine, to multiple pills, which represented a higher pill burden [15]. Polypill is not available in our setting. Third, the low level of trust that patients have in family doctors in China, as we identified in our sites [28] and another province [36], may have also contributed to the low adherence to prescriptions and healthy lifestyle advice.

We designed this pragmatic trial to be embedded in real-life primary care practice so as to better inform health policy. We minimised participant eligibility criteria to reflect the general population of patients with hypertension and/or diabetes as managed under primary care. Our data collection, based on the province-wide CVD surveillance system, enabled us to collect primary and most secondary endpoints from all participants, including patients who moved out of our study sites and those who agreed to participate but soon left the trial. We were therefore able to conduct true ITT analyses without imputation, which is not possible in many trials. The pharmaceutical and healthy lifestyle interventions were designed to fit into the job descriptions of family doctors under the new payment and incentive system of China's health reform. Our studies showed that the interventions were feasible for health providers [15] and patients [17].

Our study has several limitations. First, at recruitment all participants were verbally informed that they had 20% or greater risk of developing CVD in the next 10 years, and that they needed to adopt activities to reduce this risk, such as treating their hypertension and/or diabetes and making general healthy lifestyle changes. They all had their blood pressure measured and participated in a questionnaire survey. These activities can be seen as an intervention itself as some patients in the control arm may have modified their medicines and/or lifestyles. This potentially reduced the difference in primary and secondary outcomes observed between in the intervention and control arms. Second, we employed the Asian equation to calculate the 10-year CVD risk of participants, but the Asian equation does not consider any comorbidities or complications that may contribute to a higher risk of CVD. However, we included all patients with diabetes, who were more likely to have complications than patients with hypertension. Third, there was no blinding for healthcare providers or patients, although the outcome analyses were blinded. However, given the largely null results, there was no evidence of the typical inflation of effect sizes seen due to lack of blinding. Fourth, we could not collect baseline information for the 3,196 (10%) eligible patients who declined participation based on ethical considerations. Therefore, we could not fully assess how comparable the trial cohort was to the wider target population. Fifth, our 36-month follow-up period is shorter than that of other trials employing CVD events as the primary endpoint, which normally use 60 months. However, given the health system limitations identified here, a longer follow-up period is unlikely to generate substantially better impact on CVD events. Sixth, we were not able to collect information regarding lipid profiles, blood glucose, and healthy lifestyle behaviours for all participants due to the concern that too many data collection activities could alter routine services. We collected the information in a panel of participants and will report these results elsewhere. Seventh, recall and reporting biases may exist because we collected selfreported adherence to medicines. Electronic pill boxes were not available during our study, and even if available, they may alter provider and patient behaviour. Lastly, any attempt at robustly estimating the likely effect of the treatment if adhered to was not possible due to the very small proportion of individuals who adhered at even a modest rate in the intervention arm.

Our intervention's lack of effectiveness on CVD events is consistent with results from trials examining CVD-risk-based management interventions in primary care settings in HICs [8,37–39]. One of the major limitations of these previous trials is that their interventions were all light-touch, focusing on providing healthy lifestyle advice, with only 1 suggesting antihypertensive treatment. Unlike in these trials, we observed improved adherence to medications and significant but small reductions in blood pressure. There were another 2 trials from LMICs that implemented comprehensive pharmaceutical and lifestyle interventions among populations with high CVD risk [10,11]. Both achieved reductions in blood pressure, and one in

CVD risk, but the trials were not designed to test effects on CVD events, they had a shorter duration, and their sustainability was unclear.

CVD-risk-based pharmaceutical and healthy lifestyle management remains a popular policy for primary prevention of CVD as shown in WHO recommendations [1] and certain national guidelines [40,41]. Our study, along with other studies from HICs [37–39], showed that interventions that reduce CVD risks at the population level would be unlikely to have an impact unless structural health system factors such as the availability of essential medicines are addressed first. This calls for more implementation research to examine how interventions can work in real-world settings. The context of the health system, as revealed in our study, determines the success of implementation. Health system factors at the policy, facility, and individual level need to be considered in future research. At the policy level, improved financing of primary care is required to ensure patients have adequate access to family doctor services and essential medicines. Free distribution of essential medicines, as shown in Ontario, Canada [42], improved treatment adherence, and would be expected to reduce CVD risk factors, which in return would substantially reduce societal costs associated with poor outcomes in managing diabetes and hypertension in the long run [43]. At the facility level, risk-based management guidelines need to be made compatible with local primary care delivery to accommodate regular patient follow-ups and personalised health education [44]. At the individual level, patients have to be constantly supported and encouraged for positive changes. This requires involvement of community-based health workers, patient engagement and empowerment for disease self-management, shared decision-making, and use of telemedicine and apps in the post-coronavirus-pandemic era [2].

In conclusion, our study showed that our comprehensive package of risk-based pharmaceutical and health lifestyle interventions for patients with hypertension and diabetes improved patient adherence to essential medicines and reduced patient blood pressure, but importantly it did not reduce severe CVD events. Revisiting key health system components, such as improving universal health coverage for essential medicines, is crucial in designing future trials and policies to make an impact in risk-based primary care CVD prevention.

Supporting information

S1 CONSORT Checklist. (DOCX)

S1 Text. Supporting tables. Table A: Comparison of primary and secondary outcome disease event rates between the intervention and control arms (cluster-level data analysis). Table B: Comparison of systolic and diastolic blood pressure outcomes between the intervention and control arms (cluster-level data analysis). Table C: Comparison of time to first CVD event and time to CVD mortality outcomes between the intervention and control arms (cluster-level data analysis). Table D: Participant characteristics of per-protocol analysis. Table E: Comparison of event rates between the intervention and control groups in the per-protocol analysis. Table F: Comparison of time to first CVD event and time to CVD mortality outcomes between the intervention and control arms in the per-protocol analysis. Table G: Comparison of systolic and diastolic blood pressure outcomes between the intervention and control arms in the per-protocol analysis. Table I: Comparison of systolic and diastolic blood pressure outcomes between the intervention and control arms in the per-protocol analysis. Table I: Comparison of systolic and diastolic blood pressure outcomes between the intervention and control arms in the per-protocol analysis. Table I: Comparison of adverse events between the intervention and control arms. (DOCX)

Acknowledgments

We thank the following members in the Trial Steering Committee: Prof. Kamran Siddiqi from the University of York (Chair), Prof. Yude Chen from Peking University, Prof. Stephen Leeder from the University of Sydney, and Prof. Qiang Sun from Shandong University. We thank colleagues from Zhejiang CDC and health authorities, and those from Shaoxing Prefecture, Shangyu County, Zhuji County, and Shengzhou County, and staff in the participatory township hospitals in the 3 counties. We also thank Prof. Yunxian Yu from the School of Public Health at Zhejiang University for his input in training family doctors. We also thank all research assistants and postgraduate students from Zhejiang CDC, the Chinese University of Hong Kong, and the University of Leeds who participated in the data collection and interventions.

Transparency declaration

The lead authors affirm that the paper is an honest, accurate, and transparent account of the study being reported; that no important aspects of the study have been omitted; and that any discrepancies from the study as planned (and, if relevant, registered) have been explained.

Patient and public involvement

Patients were involved in our process evaluation and provided valuable insights regarding how the interventions were experienced by patients. Those were published in our process evaluation paper.

Dissemination declaration

We have involved health policy makers (MY, and others from the health bureaus) from the beginning in the activities including designing, piloting, and implementing the interventions. We also actively involved health practitioners in township hospitals and county hospitals for their comments and feedback. The results have been shared with health policy makers, doctors, and patients during the study, and have been used to improve hypertension and diabetes management in the research area and at the provincial level.

Author Contributions

Conceptualization: Xiaolin Wei, Guanyang Zou, John D. Walley, Min Yu.

Data curation: Zhitong Zhang, Weiwei Gong.

Formal analysis: Zhitong Zhang, Marc K. C. Chong, Joseph P. Hicks.

Funding acquisition: Xiaolin Wei, John D. Walley.

Investigation: Zhitong Zhang.

Methodology: Xiaolin Wei, Marc K. C. Chong, Joseph P. Hicks, Guanyang Zou, John D. Walley.

Project administration: Xiaolin Wei, Zhitong Zhang, Weiwei Gong, Guanyang Zou, Jieming Zhong, Min Yu.

Resources: Min Yu.

Supervision: Xiaolin Wei, Min Yu.

Validation: Xiaolin Wei, Min Yu.

Visualization: Zhitong Zhang, Marc K. C. Chong.

Writing - original draft: Xiaolin Wei.

Writing – review & editing: Xiaolin Wei, Zhitong Zhang, Marc K. C. Chong, Joseph P. Hicks, Weiwei Gong, Guanyang Zou, Jieming Zhong, John D. Walley, Ross E. G. Upshur, Min Yu.

References

- 1. World Health Organization. Technical package for cardiovascular disease management in primary health care. Geneva: World Health Organization; 2016.
- Gupta R, Wood DA. Primary prevention of ischaemic heart disease: populations, individuals, and health professionals. Lancet. 2019; 394(10199):685–96. <u>https://doi.org/10.1016/S0140-6736(19)31893-8</u> PMID: 31448740
- Geldsetzer P, Manne-Goehler J, Marcus ME, Ebert C, Zhumadilov Z, Wesseh CS, et al. The state of hypertension care in 44 low-income and middle-income countries: a cross-sectional study of nationally representative individual-level data from 1.1 million adults. Lancet. 2019; 394(10199):652–62. https:// doi.org/10.1016/S0140-6736(19)30955-9 PMID: 31327566
- NCD Risk Factor Collaboration. Long-term and recent trends in hypertension awareness, treatment, and control in 12 high-income countries: an analysis of 123 nationally representative surveys. Lancet. 2019; 394(10199):639–51. https://doi.org/10.1016/S0140-6736(19)31145-6 PMID: 31327564
- Working Group on the Summit on Combination Therapy for CVD, Yusuf S, Attaran A, Bosch J, Joseph P, Lonn E, et al. Combination pharmacotherapy to prevent cardiovascular disease: present status and challenges. Eur Heart J. 2014; 35(6):353–64. <u>https://doi.org/10.1093/eurheartj/eht407</u> PMID: 24288261
- Roshandel G, Khoshnia M, Poustchi H, Hemming K, Kamangar F, Gharavi A, et al. Effectiveness of polypill for primary and secondary prevention of cardiovascular diseases (PolyIran): a pragmatic, cluster-randomised trial. Lancet. 2019; 394(10199):672–83. https://doi.org/10.1016/S0140-6736(19)31791-X PMID: 31448738
- Alageel S, Gulliford MC, McDermott L, Wright AJ. Multiple health behaviour change interventions for primary prevention of cardiovascular disease in primary care: systematic review and meta-analysis. BMJ Open. 2017; 7(6):e015375. https://doi.org/10.1136/bmjopen-2016-015375 PMID: 28619779
- Keyserling TC, Sheridan SL, Draeger LB, Finkelstein EA, Gizlice Z, Kruger E, et al. A comparison of live counseling with a web-based lifestyle and medication intervention to reduce coronary heart disease risk: a randomized clinical trial. JAMA Intern Med. 2014; 174(7):1144–57. https://doi.org/10.1001/ jamainternmed.2014.1984 PMID: 24861959
- Lindholm LH, Ekbom T, Dash C, Isacsson A, Schersten B. Changes in cardiovascular risk factors by combined pharmacological and nonpharmacological strategies: the main results of the CELL Study. J Intern Med. 1996; 240(1):13–22. <u>https://doi.org/10.1046/j.1365-2796.1996.492831000.x</u> PMID: 8708586
- Tian M, Ajay VS, Dunzhu D, Hameed SS, Li X, Liu Z, et al. A cluster-randomized, controlled trial of a simplified multifaceted management program for individuals at high cardiovascular risk (SimCard Trial) in rural Tibet, China, and Haryana, India. Circulation. 2015; 132(9):815–24. <u>https://doi.org/10.1161/</u> CIRCULATIONAHA.115.015373 PMID: 26187183
- 11. Schwalm JD, McCready T, Lopez-Jaramillo P, Yusoff K, Attaran A, Lamelas P, et al. A communitybased comprehensive intervention to reduce cardiovascular risk in hypertension (HOPE 4): a clusterrandomised controlled trial. Lancet. 2019; 394(10205):1231–42. https://doi.org/10.1016/S0140-6736 (19)31949-X PMID: 31488369
- Karmali KN, Persell SD, Perel P, Lloyd-Jones DM, Berendsen MA, Huffman MD. Risk scoring for the primary prevention of cardiovascular disease. Cochrane Database Syst Rev. 2017; 3:CD006887. https://doi.org/10.1002/14651858.CD006887.pub4 PMID: 28290160
- Wei X, Zou G, Yin J, Walley J, Zhou B, Yu Y, et al. Characteristics of high risk people with cardiovascular disease in Chinese rural areas: clinical indictors, disease patterns and drug treatment. PLoS ONE. 2013; 8(1):e54169. https://doi.org/10.1371/journal.pone.0054169 PMID: 23349814
- Li H, Wei X, Wong MC, Yang N, Wong SY, Lao X, et al. A comparison of the quality of hypertension management in primary care between Shanghai and Shenzhen: a cohort study of 3196 patients. Medicine (Baltimore). 2015; 94(5):e455. https://doi.org/10.1097/MD.00000000000455 PMID: 25654383
- Zou G, Wei X, Gong W, Yin J, Walley J, Yu Y, et al. Evaluation of a systematic cardiovascular disease risk reduction strategy in primary healthcare: an exploratory study from Zhejiang, China. J Public Health (Oxf). 2014; 37(2):241–50. https://doi.org/10.1093/pubmed/fdu013 PMID: 24696086

- Wei X, Zou G, Gong W, Yin J, Yu Y, Walley J, et al. Cardiovascular disease risk reduction in rural China: a clustered randomized controlled trial in Zhejiang. Trials. 2013; 14:354. <u>https://doi.org/10.1186/</u> 1745-6215-14-354 PMID: 24160442
- Wei X, Walley JD, Zhang Z, Zou G, Gong W, Deng S, et al. Implementation of a comprehensive intervention for patients at high risk of cardiovascular disease in rural China: a pragmatic cluster randomized controlled trial. PLoS ONE. 2017; 12(8):e0183169. <u>https://doi.org/10.1371/journal.pone.0183169</u> PMID: 28813512
- Campbell MK, Piaggio G, Elbourne DR, Altman DG. Consort 2010 statement: extension to cluster randomised trials. BMJ. 2012; 345:e5661. https://doi.org/10.1136/bmj.e5661 PMID: 22951546
- Barzi F, Patel A, Gu D, Sritara P, Lam TH, Rodgers A, et al. Cardiovascular risk prediction tools for populations in Asia. J Epidemiol Community Health. 2007; 61(2):115–21. https://doi.org/10.1136/jech. 2005.044842 PMID: 17234869
- China National Centre for Cardiovascular Diseases. National guideline of community-based management of hypertension. Ministry of Health of the People's Republic of China; 2010.
- **21.** China Medical Association. China national type 2 diabetes mellitus prevention and control guideline. Beijing: China Medical Association; 2013.
- Lu J, Lu Y, Wang X, Li X, Linderman GC, Wu C, et al. Prevalence, awareness, treatment, and control of hypertension in China: data from 1.7 million adults in a population-based screening study (China PEACE Million Persons Project). Lancet. 2017; 390(10112):2549–58. https://doi.org/10.1016/S0140-6736(17)32478-9 PMID: 29102084
- 23. Webster R, Patel A, Selak V, Billot L, Bots ML, Brown A, et al. Effectiveness of fixed dose combination medication ('polypills') compared with usual care in patients with cardiovascular disease or at high risk: a prospective, individual patient data meta-analysis of 3140 patients in six countries. Int J Cardiol. 2016; 205:147–56. https://doi.org/10.1016/j.ijcard.2015.12.015 PMID: 26736090
- Liu L, Writing Group of 2010 Chinese Guidelines for the Management of Hypertension. [2010 Chinese guidelines for the management of hypertension.] Zhonghua Xin Xue Guan Bing Za Zhi. 2011; 39 (7):579–616. PMID: 22088239
- 25. Gong W, Wei X, Liang Y, Zou G, Hu R, Deng S, et al. Urban and rural differences of acute cardiovascular disease events: a study from the population-based real-time surveillance system in Zhejiang, China in 2012. PLoS ONE. 2016; 11(11):e0165647. <u>https://doi.org/10.1371/journal.pone.0165647</u> PMID: 27802321
- 26. Mähönen M, Cepaitis Z, Kuulasmaa K, WHO MONICA Project. Quality assessment of acute coronary care data in the WHO MONICA Project. Geneva: World Health Organization; 1999 [cited 2014 Dec 10]. Available from: http://www.thl.fi/publications/monica/accqa/accqa.htm.
- 27. Mähönen M, Tolonen H, Kuulasmaa K, WHO MONICA Project. Quality assessment of stroke event registration data in the WHO MONICA Project. Geneva: World Health Organization; 1998 [cited 2014 Apr 27]. Available from: http://www.thl.fi/publications/monica/strokega/strokega.htm.
- Zou G, Zhang W, King R, Zhang Z, Walley J, Gong W, et al. Process evaluation of a clustered randomized control trial of a comprehensive intervention to reduce the risk of cardiovascular events in primary health care in rural China. Int J Environ Res Public Health. 2020; 17(11):4156. https://doi.org/10.3390/ ijerph17114156 PMID: 32532144
- Nakagawa S, Johnson PCD, Schielzeth H. The coefficient of determination R(2) and intra-class correlation coefficient from generalized linear mixed-effects models revisited and expanded. J R Soc Interface. 2017; 14(134):20170213. https://doi.org/10.1098/rsif.2017.0213 PMID: 28904005
- Nakagawa S, Schielzeth H. Repeatability for Gaussian and non-Gaussian data: a practical guide for biologists. Biol Rev Camb Philos Soc. 2010; 85(4):935–56. https://doi.org/10.1111/j.1469-185X.2010. 00141.x PMID: 20569253
- Colhoun HM, Betteridge DJ, Durrington PN, Hitman GA, Neil HAW, Livingstone SJ, et al. Primary prevention of cardiovascular disease with atorvastatin in type 2 diabetes in the Collaborative Atorvastatin Diabetes Study (CARDS): multicentre randomised placebo-controlled trial. Lancet. 2004; 364 (9435):685–96. https://doi.org/10.1016/S0140-6736(04)16895-5 PMID: 15325833
- **32.** Burnier M, Egan BM. Adherence in hypertension. Circ Res. 2019; 124(7):1124–40. https://doi.org/10. 1161/CIRCRESAHA.118.313220 PMID: 30920917
- Zou G, Zhang Z, Walley J, Gong W, Yu Y, Hu R, et al. Use of medications and lifestyles of hypertensive patients with high risk of cardiovascular disease in rural China. PLoS ONE. 2015; 10(5):e0124484. https://doi.org/10.1371/journal.pone.0124484 PMID: 25932640
- Fang H, Eggleston K, Hanson K, Wu M. Enhancing financial protection under China's social health insurance to achieve universal health coverage. BMJ. 2019; 365:I2378. https://doi.org/10.1136/bmj. I2378 PMID: 31227485

- Jing Z, Chu J, Imam Syeda Z, Zhang X, Xu Q, Sun L, et al. Catastrophic health expenditure among type 2 diabetes mellitus patients: a province-wide study in Shandong, China. J Diabetes Investig. 2019; 10 (2):283–9. https://doi.org/10.1111/jdi.12901 PMID: 30044060
- 36. Wei X, Deng S, Haldane V, Blacklock C, Zhang W, Zhang Z, et al. Understanding factors influencing antibiotic prescribing behaviour in rural China: a qualitative process evaluation of a cluster randomized controlled trial. J Health Serv Res Policy. 2020; 25(2):94–103. <u>https://doi.org/10.1177/ 1355819619896588</u> PMID: 31986910
- Wilhelmsen L, Berglund G, Elmfeldt D, Tibblin G, Wedel H, Pennert K, et al. The multifactor primary prevention trial in Goteborg, Sweden. Eur Heart J. 1986; 7(4):279–88. <u>https://doi.org/10.1093/</u> oxfordjournals.eurheartj.a062065 PMID: 3720755
- Holt TA, Thorogood M, Griffiths F, Munday S, Friede T, Stables D. Automated electronic reminders to facilitate primary cardiovascular disease prevention: randomised controlled trial. Br J Gen Pract. 2010; 60(573):e137–43. https://doi.org/10.3399/bjgp10X483904 PMID: 20353659
- Jorgensen T, Jacobsen RK, Toft U, Aadahl M, Glumer C, Pisinger C. Effect of screening and lifestyle counselling on incidence of ischaemic heart disease in general population: Inter99 randomised trial. BMJ. 2014; 348:g3617. https://doi.org/10.1136/bmj.g3617 PMID: 24912589
- 40. Harris MF, Lloyd J, Litt J, van Driel M, Mazza D, Russell G, et al. Preventive evidence into practice (PEP) study: implementation of guidelines to prevent primary vascular disease in general practice protocol for a cluster randomised controlled trial. Implement Sci. 2013; 8:8. https://doi.org/10.1186/1748-5908-8-8 PMID: 23327664
- Pylypchuk R, Wells S, Kerr A, Poppe K, Riddell T, Harwood M, et al. Cardiovascular disease risk prediction equations in 400 000 primary care patients in New Zealand: a derivation and validation study. Lancet. 2018; 391(10133):1897–907. https://doi.org/10.1016/S0140-6736(18)30664-0 PMID: 29735391
- Persaud N, Bedard M, Boozary AS, Glazier RH, Gomes T, Hwang SW, et al. Effect on treatment adherence of distributing essential medicines at no charge: the CLEAN Meds Randomized Clinical Trial. JAMA Intern Med. 2019; 180(1):27–34. <u>https://doi.org/10.1001/jamainternmed.2019.4472</u> PMID: 31589276
- 43. Wirtz VJ, Kaplan WA, Kwan GF, Laing RO. Access to medications for cardiovascular diseases in lowand middle-income countries. Circulation. 2016; 133(21):2076–85. https://doi.org/10.1161/ CIRCULATIONAHA.115.008722 PMID: 27217433
- Russell G, Lane R, Parker S, Litt J, Mazza D, Lloyd J, et al. Preventive evidence into practice: what factors matter in a facilitation intervention to prevent vascular disease in family practice? BMC Fam Pract. 2019; 20(1):113. https://doi.org/10.1186/s12875-019-0995-7 PMID: 31395020